Author's response to reviews

Title: Neonatal ECMO Study of Temperature (NEST) - a randomised controlled trial

Authors:

David J Field Prof (df63@le.ac.uk)
Richard Firmin Mr (richard.firmin@uhl-tr.nhs.uk)
Denis V Azzopardi Dr (d.azzopardi@imperial.ac.uk)
Frances Cowan Dr (f.cowan@imperial.ac.uk)
Edmund Juszczak Mr (ed.juszczak@npeu.ox.ac.uk)
Peter Brocklehurst Prof (peter.brocklehurst@npeu.ox.ac.uk)

Version: 4 Date: 8 April 2010

Author's response to reviews:

Dear Sir
This revision deals with formatting changes you requested to:
The author list;
Abstract format;
Table 1;
Removal of "protocol" from the title page;
Placement of funding statement.
These have all been completed.
Some requests could not be met eg remove qualifications from authors but in fact none had been included. I was also not clear about comments regarding author groups.
Please let me know if further changes are required.
Regards
David Field